\documentclass[10pt,letterpaper, final]{report} % \documentclass{} required in all files

\usepackage{../preamble/myWEPS} % This is now our single preamble file inclusion (it assumes a “.sty” file extension)


\begin{document}
The desired soil can be selected and accessed for editing from the soil input panel on the bottom of the main WEPS screen (\cref{fig:WEPS_main_screen})
and is shown here (\cref{fig:Soil_Input_panel}).

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_ref/Soil_Input_panel.jpg}
\caption{Soil input panel at the bottom of the WEPS main screen.}
\label{fig:Soil_Input_panel}
\end{figure}

Soil files can be selected from a drop-down choice list (\cref{fig:Soil_choice_list})
by using the down arrow box
\img{images/icons/down_arrow.png}
on the far right side of the bottom "Soil" panel of the WEPS main screen (\cref{fig:WEPS_main_screen}).
Navigation is performed by clicking on the ‘closed key’ symbol
\img{images/icons/ssurgoopenkey.png}
or select the folder and press the right arrow key to open the folder to display its contents.
To close a folder, click the ‘open key’ symbol
\img{images/icons/ssurgoclosekey.png}
or press the left arrow key 
to quit displaying the folder contents.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_Ref/Soil_choice_list.jpg}
\caption{Soil selection drop down choice list.}
\label{fig:Soil_choice_list}
\end{figure} 

\hl{The section below is not working 100 percent as documented yet: Mantis ticket 3791: Soil "P" and "T" button FileChooser updates.}
The user can also select a soil by clicking on the \textbf{Template} folder
\img{images/icons/foldert.png}
to the left of the soil input field, which will bring up a standard file chooser window (\cref{fig:Template_Soil_FileChooser}).
Similarly, the user may also select previously soil files stored in the current WEPS Project directory by clicking on the \textbf{Project} folder
\img{images/icons/folderp.png}
button.

\begin{figure}[H]
\centering
	\includegraphics[scale=1.75]{images/Interface_ref/Template_Soil_FileChooser.jpg}
\caption{Template soil file chooser window.}
\label{fig:Template_Soil_FileChooser}
\end{figure}

Soil files in a WEPS Project directory are usually copies of previously selected soil files
obtained from the NRCS NASIS (SSURGO) soil database.
Note that the
\img{images/icons/folderp.png}
and
\img{images/icons/foldert.png}
buttons are not displayed by default for the official NRCS configured releases of WEPS.
Therefore, these users can only use the down arrow box
\img{images/icons/down_arrow.png}
to access the drop-down choice list for selecting soil data from the main WEPS screen.

% \begin{figure}[H]
% \centering
% 	\includegraphics[scale=0.80]{images/Interface_ref/ssurgoselectNew.jpg}
% \caption{The “Select Soil” window.}
% \label{fig:Select_Soil}
% \end{figure}

%A soil for a WEPS simulation run is typically selected from a list of soils contained in the NRCS Soil Survey Geographic (SSURGO) database
%via a live \textbf{NRCS Soil DataMart} internet connection option.
%There are other options available though, through the \textbf{Soil Templates} subfolder with \textbf{Exercise Soils},
%\textbf{NRCS Generic Soils} and optionally Microsoft Access NRCS SURRGO “.mdb” files.
%Note that NRCS and other users can also provide MS Access SSURGO files for use in offline (no internet access) conditions.
%In addition, the current WEPS \textbf{Project} folder containing previously selected “.ifc” soil files is also available.

%Clicking on the ‘Template’ folder opens a window titled ‘Select Soil’ (\cref{fig:Template_Soil_FileChooser}).
%Navigate through the database tree to find the soil survey area (or county) desired.
%Navigation is performed by clicking on the ‘closed key’ symbol
% \img{images/icons/ssurgoopenkey.png}
%to open the folder to display its contents.
%To close a folder, click the ‘open key’ symbol
% \img{images/icons/ssurgoclosekey.png}
%to quit displaying the folder contents.
%The soil files are listed according to the soil map unit symbol, map unit name, surface texture, and local phase.
%Selecting a soil then displays its components and the percentage that each component contributes to the map unit.
%Click a soil component to highlight it and click the ‘Select’ button
% \scalerel*{\img{images/icons/select.png}}{\strut}
%at the bottom of the screen (or double click the component with the left mouse button).
%This action converts the soil from the SSURGO database to a WEPS soil file format (with an ‘ifc’ file extension),
%places a copy in the current “Project” folder and returns the user to the main screen.
%The loaded soil file name will appear in the soil box window. Clicking the \textbf{‘Cancel’}
% \scalerel*{\img{images/icons/cancel.png}}{\strut}
%button in the \textbf{‘Select Soil’} window aborts the selection of a new soil. 
%
Soils that have been previously selected or modified and saved to another name (and stored in the current Project folder)
can be opened by clicking on the \textbf{Project} folder
\img{images/icons/folderp.png}
button.
This will open a soil filechooser window in which the user can select the desired soil or type in the soil file name. 

Users have the option to view the parameters for the soil file displayed in the soil box by clicking on the button labeled \textbf{Soil}
on the left side of the soil box.
This will open the WEPS Soil User Interface screen (\cref{fig:Soil_Interface})
which allows the user to view, edit, and save soil and initial condition information.
Details on editing a soil ".ifc" formatted file are discussed here: \hl{Editing with the Soil Interface window}.

\begin{figure}[H]
\centering
	\includegraphics[scale=0.72]{images/Interface_ref/ssurgoimportNew.jpg}
\caption{WEPS Soil User Interface screen.}
\label{fig:Soil_Interface}
\end{figure}

\end{document}
